Author’s response to reviews

Title: Debate: Why should gender-affirming health care be included in health science curricula?

Authors:

Elma de Vries (devrieselma@gmail.com;elma.devries@uct.ac.za;elma.devries@westerncape.gov.za)

Alex Müller (alex.muller@uct.ac.za)

Harsha Kathard (harsha.kathard@uct.ac.za)

Version: 1 Date: 04 Jun 2019

Author’s response to reviews:

Thank you for the opportunity to resubmit. The wording has been changed where it was similar to other publications, and direct quotes have been placed in inverted commas.